Author's response to reviews

Title: Preliminary study of relationships between hypnotic susceptibility and personality disorder functioning styles in healthy volunteers and personality disorder patients

Authors:

Fenghua Wang (7867373@qq.com)
Wanzhen Chen (cwzydzyzi@126.com)
Jingyi Huang (shizuka_11@163.com)
Peiwei Xu (drwangwei@zju.edu.cn)
Wei He (prettyhehe8899@163.com)
Hao Chai (chaihao2009@yahoo.cn)
Junpeng Zhu (zhujunpeng_007@163.com)
Wenjun Yu (selinayu@zju.edu.cn)
Li Chen (ziwfh@hotmail.com)
Wei Wang (drrwang@doctor.com)

Version: 3 Date: 28 July 2011

Author's response to reviews:

Dear Sirs:

Please see our point-to-point answers below:

Major revisions (we require the author to make these changes). Structure: Please include a conclusion after the discussion.

[our answer: Yes, we have included a Conclusion after Discussion, please see page 11]

Authors' contributions: Please indicate that all authors read and approved the final manuscript.

[Our answer: Yes, all authors have read and approved the final version of the paper]

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove from the system.

[Our answer: Yes, we have removed the highlighting traces]

Please remove the wording at the top of the page, other than the title.
[Our answer: Yes, we have removed the wording at the top of every page]

Corresponding author: Please provide email address for both corresponding authors.

[Our answer: Yes, emails have been added for both corresponding authors, please see page 1]

Structure: Change introduction to background.

[Our answer: Yes, we have changed Introduction into Background, please see Page 3]

Best,

Yours,

Wei